Author’s response to reviews

Title: Plasma HER2ECD a promising test for patient prognosis and prediction of response in HER2 positive breast cancer: Results of a randomized study - SAKK 22/99

Authors:

Serenella Eppenberger-Castori (sere.eppenberger@unibas.ch)
Dirk Klingbiel (dirk.klingbiel@roche.com)
Thomas Ruhstaller (thomas.ruustaller@kssg.ch)
Daniel Dietrich (daniel.dietrich@sakk.ch)
Daniel Rufle (danielalexander.rufle@aprentas.com)
Karin Rothgiesser (karin.Rothgiesser@sakk.ch)
Olivia Pagani (opagani@bluewin.ch)
Beat Thürlimann (beat.thuerlimann@kssg.ch)

Version: 3 Date: 24 Jan 2020

Author’s response to reviews:

Dear Editors

Thank you so much for having given us the opportunity of revising our manuscript.

In the present version

1. We explain to have received the permission to publish from the SAKK board.

2. Figures
   All figures were renamed and included in the manuscript.

3. Abbreviations
   A list of all the abbreviations used in the manuscript is provided and placed before the Declarations section.
4. Clean manuscript
A single complete and clean manuscript is upload

Looking forward to receiving the final approval

With best regards

Serenella Eppenberger